Author's response to reviews

Title: Gender-specific association between hyperuricemia and atrial fibrillation in rural China: a cross-sectional study

Authors:

Guozhe Sun (gzhsun66@163.com)
Liang Guo (cmu1h_gl@126.com)
Jun Wang (cmu_wj@126.com)
Ning Ye (653984551@qq.com)
Xunzhang Wang (Xunzhang.Wang@cshs.org)
Yingxian Sun (cmu1h_syx@126.com)

Version: 2 Date: 14 December 2014

Author's response to reviews: see over
Dear Editor,

Thank you very much for your advice for our manuscript entitled “Gender-specific association between hyperuricemia and atrial fibrillation in rural China: a cross-sectional study”. We have revised the manuscript as followings:

(1) **Title Page**: We add the email addresses of all authors in the title page.

(2) **Line Numbering**: We add line numbers in the text of our manuscript.

(3) **Authors’ Contributions**: We add the authors' contributions section in the end of our manuscript before the section of “References”.

(4) **Availability of Supporting Data section**: All the data in our manuscript are collected during our survey on each participant and no other supporting database is involved.

(5) **Deposition of data**: It is a cross-sectional population-based study and no related database is involved in our study.

Yours sincerely,

Ying-Xian Sun

Corresponding author

E-mail: cmu1h_syx@126.com